23782242, 0, Downloaded from https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/lol2.10286 by HGF GEOMAR Helmholtz Centre of Ocean, Wiley Online Library on [01/11/2022]. See the Terms

and Conditions (https://onlinelibrary.wiley

nditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons



Open Access

© 2022 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.10286

CURRENT EVIDENCE

Environmental variability in aquatic ecosystems: Avenues for future multifactorial experiments

Miriam Gerhard ,^{1*} Apostolos-Manuel Koussoroplis, ², ² Michael Raatz ³, ³ Christian Pansch, ⁴ Samuel B. Fey, ⁵ Jahangir Vajedsamiei, ⁶ Maria Calderó-Pascual, ⁷ David Cunillera-Montcusí, ^{8,9} Noël P. D. Juvigny-Khenafou ¹⁰, ¹⁰ Francesco Polazzo, ¹¹ Patrick K. Thomas ¹, ¹ Celia C. Symons ¹, ¹² Meryem Beklioğlu, ¹³ Stella A. Berger, ¹⁴ Rosa M. Chefaoui, ¹⁵ Kemal Ali Ger, ¹⁶ Silke Langenheder ¹⁷, ¹⁷ Jens C. Nejstgaard, ¹⁴ Robert Ptacnik, ⁸ Maren Striebel ¹

¹Institute for Chemistry and Biology of the Marine Environment, University of Oldenburg, Wilhelmshaven, Germany; ²Université Clermont Auvergne, CNRS, Laboratoire Microorganismes Génome et Environnement, Clermont-Ferrand, France; ³Department for Evolutionary Theory, Max Planck Institute for Evolutionary Biology, Plön, Germany; ⁴Environmental and Marine Biology, Åbo Akademi University, Turku, Finland; ⁵Department of Biology, Reed College, Portland, Oregon; ⁶Department of Marine Ecology, GEOMAR Helmholtz Centre for Ocean Research, Kiel, Germany; ⁷Centre for Freshwater and Environmental Studies, Dundalk Institute of Technology, Dundalk, Ireland; ⁸AquaScale Lab, WasserCluster Lunz - Biologische Station GmbH, Lunz am See, Austria; ⁹Freshwater Ecology, Hydrology and Management Group (FEHM), Section of Ecology, Department of Evolutionary Biology, Ecology and Environmental Sciences, University of Barcelona, Barcelona, Spain; ¹⁰iES-Institute for Environmental Sciences, University of Koblenz-Landau, Germany; ¹¹Department of Ecotoxicology, IMDEA Water Institute, Alcalá de Henares, Spain; ¹²Ecology and Evolutionary Biology, University of California, Irvine, California; ¹³Department of Biological Sciences and Centre for Ecosystem Research and Implementation, Middle East Technical University, Ankara, Turkey; ¹⁴Department of Plankton and Microbial Ecology, Leibniz Institute of Freshwater Ecology and Inland Fisheries, Stechlin, Germany; ¹⁵Area of Biodiversity and Conservation, E.S.C.E.T. Universidad Rey Juan Carlos, Madrid, Spain; ¹⁶Department of Ecology, Federal University of Rio Grande do Norte, Natal, Rio Grande do Norte, Brazil; ¹⁷Department of Ecology and Genetics/Limnology, Uppsala University, Uppsala, Sweden

Scientific Significance Statement

Variability is inherent to all natural ecosystems, yet the consequences of alterations to existing variability patterns in environmental factors expected under global change scenarios remain unclear. Motivated by observed and predicted changes, investigations including or focusing on variability are accumulating. However, no common framework exists for researching variability of single and multiple environmental factors, and mismatches between theory and experimental data at different levels challenge our current understanding of the role variability plays in nature. We identify sources of mismatches,

^{*}Correspondence: miriam.libertad.gerhard@uni-oldenburg.de

Author Contribution Statement: MG, MS, and AMK discussed the concept and frame of the paper. MG prepared a first draft of the manuscript and coordinated the discussion rounds. MG, MS, AMK, MR, CP, SBF, JV, MCP, DCM, FP, PKT, CCS, and SAB provided pieces of text or collaborated with the figures. MG, MS, AMK, MR, CP, SBF, JV, MCP, DCM, FP, PKT, CCS, and JCN participated in the discussion rounds (ideas development) of different topics. All authors revised, commented, and edited the manuscript during multiple revision rounds and approved the final version for publication.

Data Availability Statement: Data of the survey are available in the GitHub/Zenodo repository at https://zenodo.org/badge/latestdoi/460100099 (Gerhard et al. 2022).

Associate editor: Monika Winder

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

23782242, 0, Downloaded from https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/lo12.10286 by HGF GEOMAR Helmholtz Centre of Ocean, Wiley Online Library on [01/11/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/term and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

challenges, and knowledge gaps to contribute to a research agenda on the effects of variability in aquatic systems. We also provide guidance on how to investigate variability effects experimentally, considering modeling and experimental design.

Abstract

The relevance of considering environmental variability for understanding and predicting biological responses to environmental changes has resulted in a recent surge in variability-focused ecological research. However, integration of findings that emerge across studies and identification of remaining knowledge gaps in aquatic ecosystems remain critical. Here, we address these aspects by: (1) summarizing relevant terms of variability research including the components (characteristics) of variability and key interactions when considering multiple environmental factors; (2) identifying conceptual frameworks for understanding the consequences of environmental variability in single and multifactorial scenarios; (3) highlighting challenges for bridging theoretical and experimental studies involving transitioning from simple to more complex scenarios; (4) proposing improved approaches to overcome current mismatches between theoretical predictions and experimental observations; and (5) providing a guide for designing integrated experiments across multiple scales, degrees of control, and complexity in light of their specific strengths and limitations.

Organisms inhabiting aquatic ecosystems regularly experience temporal variability in multiple environmental factors simultaneously. In contrast, researchers tend to disproportionately utilize temporally static models and experiments, raising concerns for accurately predicting ecological responses amidst altered environmental variability accompanying global change (Wang and Dillon 2014; Zhang et al. 2016). Thus, the consequences of variability in environmental factors, involving single and multiple factors, have received an increased emphasis in recent empirical and theoretical studies (Gunderson et al. 2016; Koussoroplis et al. 2017; Safaie et al. 2018; Kapsenberg and Cyronak 2019; Ryo et al. 2019; Kroeker et al. 2020; Jackson et al. 2021; Pansch et al. 2022). This growing body of research accompanies increased complexity in designing and conducting experimental studies, and in interpreting the implications of the results. Variability research currently lacks an integrated framework for explicitly connecting key questions related to environmental variability with specific experimental approaches, especially those aimed at understanding the consequences of concurrent changes in the variability of multiple environmental factors. While synthesis literature is available regarding variability in terrestrial systems (Colinet et al. 2015, 2018) fewer efforts have been made to review environmental variability knowledge and incorporate it into experimental designs for aquatic ecosystems.

Here, we synthesize relevant terms and frameworks for single and multifactorial variability research and identify major research challenges in recent experimental work. Finally, we outline a path forward by discussing the integration of theoretical and experimental design aspects in aquatic system research. Although this work is mainly focused on experimental approaches in environmental variability research, this should be done in conjunction with theoretical and field approaches, as shown in this synthesis. We focus on the temporal variability of abiotic environmental drivers; however, as mobile organisms can experience temporal variability due to existing spatial variability, we extend the scope of this contribution to additionally consider the spatial domain. This manuscript focuses on aquatic ecosystems, especially regarding the experimental aspects, but general variability background and terrestrial experimental examples are included for comparison and to fill research gaps.

Relevant frameworks

Terms and definitions

Environmental variability characterizes natural ecosystems as having either deterministic (predictable, e.g., daily and seasonal light, pH, and temperature cycles) or stochastic (unpredictable, e.g., changes in estuarine and stream salinities) fluctuations (Fujiwara and Takada 2017; Kapsenberg and Cyronak 2019; Pansch and Hiebenthal 2019; Dobry et al. 2021). Natural variability in the same factor, in turn, may be perceived by organisms in a variety of ways and scales. The performance of photosynthetic organisms in aquatic systems, for instance, can be affected by light variability, such as changes in light intensity and spectrum, as seasonal or daily sinusoidal fluctuations, or rapid fluctuations due to vertical mixing of the water column, movement of organisms, and clouds (Shatwell et al. 2012; Hintz et al. 2022; Neun et al. 2022). Some environmental factors are not only influenced by abiotic but also by biotic processes. Water pH can show relevant daily fluctuations associated with photosynthesis rates in habitats with microalgae or seagrass dominance as well as stochastic variations associated with freshwater inputs (Rivest et al. 2017; Wahl et al. 2018; Cyronak et al. 2020). Similarly, oxygen and carbon dioxide availability are affected by temperature (gas solubility and water level stratification) and by the balance of respiration and photosynthesis rates (Flanagan and McCauley 2010;

Yvon-Durocher et al. 2010; Roman et al. 2019; Rasmusson et al. 2020; Wahl et al. 2021). These examples show how natural systems comprise multiple factors varying simultaneously in different manners (i.e., at different scales), generating natural complex patterns of variability.

Anthropogenic activities can also generate, exacerbate or modify variability patterns in ecosystems, increasing stressful conditions for organisms (e.g., heatwaves, pollutant concentrations) (Striebel et al. 2016; Pansch et al. 2018; Woolway et al. 2021; Polazzo et al. 2022). Novel environmental variability produces effects at different temporal and spatial scales (Boyd et al. 2016; Kroeker et al. 2020), by affecting directly the performance of the organisms, or indirectly through species interactions (e.g., affecting consumers by changes in producers) (Berger et al. 2014; Litchman et al. 2015; Harris et al. 2018). In addition, global changes caused by anthropogenic activities such as continuous press disturbances may impact differently on organisms that experience naturally variable or more constant environments. It has been proposed that warming effects on organisms' performance can be exacerbated by thermal fluctuations in comparison to warming effects under constant conditions (Wang et al. 2019: Cabrerizo et al. 2021: González-Olalla et al. 2022). However, water pH fluctuations under ocean acidification scenarios show less conclusive results, where negative, neutral, or positive effects on performance have been detected for corals and coralline algae when exposed to fluctuations under low pH (Cornwall et al. 2013, 2018; Rivest et al. 2017). Such results highlight the relevance of considering natural variability when attempting to predict biotic responses to global change and the need for additional research examining a broad range of variables.

Natural and anthropogenic-induced dynamics of variability can thus generate a wide range of environmental variability patterns described by different characteristics (i.e., components), including the magnitude, frequency, and predictability of variation (Box 1). Such components are commonly assessed in variability research, depending on the investigation scale (e.g., single or multiple events, see Ryo et al. 2019 for synthesis in temporal dynamics), but ecologists still lack a common framework. Establishing consensus on how to define variability components and how to analyze their effects in a multifactorial context is not trivial and doing so would facilitate the query for specific research directions, enhance communication among researchers, and allow the comparison of outcomes. Thus, the first step for a variability research framework necessitates producing common definitions for the sources of variability, the components describing variability, and the types of variability effects in a multifactorial context (summarized in Box 1).

How to approach biological consequences of environmental variability

Our current experimental knowledge on biological responses (e.g., traits) to environmental factors is primarily generated in constant laboratory environments with a focus on changes in mean values, or as ramping assays (gradual change) in longer-lived organisms (Rezende et al. 2014). These responses can often be modeled as nonlinear functions of the environmental gradient (Angilletta 2006; Denny 2017). In a variable environment, however, plugging the average value of the environment in the nonlinear function can approximate, or either underestimate or overestimate the observed response (Niehaus et al. 2012). Such bias can result from Jensen's inequality (Jensen 1906), a mathematical property that implies different biological responses between variable and constant environments even if the two environments share the same average conditions (Hastings and Caswell 1979; Ruel and Avres 1999). This property predicts that for convex (i.e., accelerating) regions of response functions, environmental variability increases the mean biological responses relative to constant environmental regimes with similar mean conditions, whereas for concave (i.e., decelerating) regions of response functions, variability effects are reversed (Ruel and Ayres 1999). To predict how much biological responses differ, that is, the variance effect, one can use statistical methods such as those developed in the Scale Transition Theory (Chesson et al. 2005). The magnitude of the increase or decrease of the response is defined by the function's curvature and the environmental factor's statistical variance. The Scale Transition Theory is now increasingly used in ecology to formulate and test hypotheses of thermal fluctuation effects on vital rates at different levels of biological organization from performance curves obtained under constant or static conditions in experimental (Kingsolver and Woods 2016; Bernhardt et al. 2018; Gerhard et al. 2019) and theoretical (Vasseur et al. 2014; Dowd et al. 2015; Denny 2017, 2019) studies. For example, Bernhardt et al. (2018) confirmed the above predictions experimentally by comparing phytoplankton thermal performance curves under constant and variable thermal regimes, where observed performance under variable conditions matched the predicted values for the convex and concave parts of the curve measured under constant conditions (see also Morash et al. 2018 as an example on fish). Despite the Scale Transition Theory predictions are theoretically valid for other environmental variables than temperature (Denny and Dowd 2022), which show nonlinear performance curves (e.g., resource supply, Litchman 2000; Bestion et al. 2018), this approach has been less used for testing performance under constant and fluctuating conditions in other factors (see Section Challenges of integrating inferences from experimental data and theory on effects of environmental variability).

Scale Transition Theory is a valuable tool for understanding the effects of variability and for stating qualitative hypotheses, but like any statistical method, it has limitations. First, mathematically, the Scale Transition Theory equations are exact only in the case of precisely quadratic biological response functions. In all other cases, the equations are reasonable approximations as long as environmental variability remains relatively low. Second, even when the predictions are

Box 1. Important terms (definitions) in multifactorial environmental variability ecology.

Factor identity

Environmental factor(s) that are considered (measured or manipulated), for example, nutrient and pollutant concentrations, temperature, light spectrum and intensity, oxygen concentration, salinity, *p*CO2/pH, hydrodynamic patterns. Such environmental factors have direct effects on organisms and indirect effects by altering interactions. The relevance of different factors might differ across systems and sites (e.g., within and between ponds, lakes, rivers, and marine systems).

Scale

- Depending on the target organizational level and organism life span, variability is perceived differently (e.g., as continuous or fluctuating) and therefore different scales may have to be considered (Petersen et al. 2009; Boyd et al. 2016; Gunderson et al. 2016; Jackson et al. 2016; Kroeker et al. 2020)
- Temporal: From minutes to subdaily to decadal timescales. Temporal scales are related to the duration of environmental events and biological processes (e.g., acute vs. chronic responses, organism's lifespan).
- Spatial: From nm–μm to km across latitudinal and longitudinal gradients. In aquatic ecosystems the spatial scale can be described by different variables such as length, depth, area, volume, shape, and habitat heterogeneity.

Components of variability

There is a wide range of characteristics of environmental variability drivers used for describing natural dynamics or novel patterns. Patterns of variability may become stressors in disturbed ecosystems. Such characteristics are frequently used in variability research depending on the investigation scale but lack a common language and definitions. Different metrics are commonly used for describing variability according to the scale of evaluation (Ryo et al. 2019):

Short scales/single events

- Magnitude: Is the amount of change in one factor and characterizes the intensity of one or more events (e.g., the amplitude of daily temperature fluctuations).
- Rate of change: Indicates if the experienced variability is abrupt or gradual.
- Duration: Is the length of a specific event (Coble et al. 2016). For example, the time between start and end (dates) of heatwaves (Oliver et al. 2018).

Large scale

- Variance: Measures the distribution of an environmental variable around its mean (Coble et al. 2016) and provides information about deviation from mean values at different scales of interest. The amplitude is also commonly used to approximate the variance of deterministic patterns of variability (e.g., diurnal cycles).
- Frequency: Is the number of regular cycles or events that occur within a given time period. Can be analyzed using spectral analysis and Fourier transformation (Dillon et al. 2016; Kroeker et al. 2020). The same concept can be used for spatial variation and is referred to as the grain size of environmental patches (Koussoroplis et al. 2019).
- Autocorrelation: This aspect has been analyzed as the color of noise (Travis 2001; Vasseur and Yodzis 2004), where variability is
 defined as white noise when no autocorrelation and constant variance are present in any scale. Red noise, in turn, has stronger
 autocorrelation and increasing variance with time and/or space. A higher autocorrelation usually implies higher predictability.
 Predictability has been also considered as the consistency in the magnitude and timing of environmental fluctuations (Kroeker
 et al. 2020).

Multifactorial context

When more than one factor is evaluated, there are different aspects and effects to consider.

Combined (interactive) effects:

- *Additive*: When the response to the combination of more than one factor together *equals* the sum of the responses of the single factors' addition.
- *Nonadditive (or interactive)*: When the response to the combination of more than one factor together *differs* from the sum of the responses of the single factors' addition.
 - *Synergistic*: When the response to addition of more than one factor together is *larger* than the sum of the responses of the single factors' addition (Koussoroplis et al. 2017).
 - *Antagonistic*: When the response to addition of more than one factor together is *smaller* than the sum of the responses of the single factors' addition (Koussoroplis et al. 2017).

Special cases of nonadditivity:

- Cross-dependence: The influence of the level of one factor on the system response shape (response curve) to another factor (Fig. 2, Koussoroplis et al. 2017).
- Covariance effect: The sensitivity of the system response to the covariance of multiple factors (Koussoroplis et al. 2017). Environmental factors can show positive (in-phase) or negative (out-of-phase) covariation dynamics (Fig. 2).

mathematically exact, Scale Transition Theory predictions can differ from observations indicating that other mechanisms besides nonlinearity are in play (Niehaus et al. 2012; Koussoroplis et al. 2017). Variability may induce compensatory acclimation or stress that changes patterns of retrieved performance functions (Vajedsamiei et al. 2021b). Gradual plasticity (i.e., progressive phenotypic adjustments by individual organisms or populations in response to environmental change) amidst environmental fluctuations can generate mismatches between the realized performance and expected responses caused by the delay in phenotypic adjustments relative to environmental change (Kremer et al. 2018; Fey et al. 2021). This implies that realized performance depends not only on the variance of environmental factors but also on the temporal scale at which the factors fluctuate, a phenomenon termed time-dependent effects (Rezende et al. 2014; Dowd et al. 2015; Kingsolver and Woods 2016; Koussoroplis et al. 2017; Vajedsamiei et al. 2021b). Time-dependent effects have been experimentally studied for temperature variability, demonstrating how increased duration of exposure to stressful temperatures can narrow organismal thermal tolerances (Rezende et al. 2014), and how daily fluctuations can exacerbate acute responses to high temperatures (Kingsolver et al. 2015). Thus, gradual plasticity can enhance or decrease realized performance compared to expectations for thermal variation depending on the thermal acclimation history, as has been demonstrated experimentally (Kremer et al. 2018; Fey et al. 2021; Vajedsamiei et al. 2021b). Most information for building this framework is based on temperature studies, however, studies involving other environmental variables also support these ideas. Under slow light fluctuations, phytoplankton has been shown to integrate growth rates from extreme values by adjusting to changes in light intensity, while fast fluctuations reflect the average growth rates of constant conditions (Litchman 2000). Furthermore, simulations of salinity reduction caused by storms showed that not only the salinity level (i.e., magnitude), but the length and timing of this event alters the growth of a marine snail larvae (Richmond and Woodin 1996). Here, larvae exposed to later, longer or lower salinity events presented smaller size after the salinity manipulations (storms) were ended (Richmond and Woodin 1996).

Integrating multifactorial variability

The performance of biological systems is rarely determined by a single environmental factor (Vinebrooke et al. 2004; Boyd et al. 2016). Co-occurring factors may combine and lead to different effects than those arising from the two factors in isolation, driven by synergistic or antagonistic effects (Box 1, Gunderson et al. 2016; Jackson et al. 2016). Such interactions among factors can be highly complex since the curvature of the response to each factor (when considering a gradient) as well as the joint factor effect (additive, nonadditive) can change for different conditions (cross-dependence, Sperfeld et al. 2016; Koussoroplis et al. 2017). The effect of temperature and nutrients on primary producers is a well-studied example of complex interactions. Experiments show that high temperature increases phytoplankton biomass when nutrient supply is high, but the temperature effect disappears or even turns negative under low nutrient regimes (Hennemann and Petrucio 2010; De Senerpont Domis et al. 2014; Verbeek et al. 2018). When extending the levels of each factor to gradients, nutrient availability was shown to change phytoplankton's thermal performance curves by changing the thermal optima and breadth (Thomas et al. 2017; Bestion et al. 2018; Aranguren-Gassis et al. 2019). Thus, the effect size of the thermal variance changes in accordance with how nutrients shape thermal performance curves (Koussoroplis et al. 2017; Gerhard et al. 2019). More generally, this means that if interactive effects change along the environmental gradient, sensitivity to variance might change in effect size and even direction (Sperfeld et al. 2016).

Scale Transition Theory equations can be readily expanded to multivariate response functions, and provide important insights regarding the link between the way factors jointly act upon the biological response and their variance effects (Chesson et al. 2005; Denny 2016; Koussoroplis and Wacker 2016; Koussoroplis et al. 2017). If the effect of multiple factors on a biological response is additive, the effect of their variances is also additive, and the factors can be considered separately. However, when the effect of two environmental factors on a biological system is nonadditive (presence of interactive effects), the biological response also depends on the statistical covariance between the factors, that is, the covariance effect (Denny 2016; Koussoroplis and Wacker 2016; Koussoroplis et al. 2017, 2019). Theory predicts that covariance effects are proportional to the covariance between the factors and the degree of nonadditivity (i.e., the magnitude of the response to the combination of two factors compared to the sum of the effects of each factor alone). The direction of the covariance effect changes with the nonadditivity type (antagonistic or synergistic). For example, if two factors covary positively and their effect is synergistic the covariance effect will tend to increase the biological response, but the response will tend to decrease if the two factors act antagonistically (Koussoroplis et al. 2017). Given the high relevance of nonadditive effects on organism's performance (Gunderson et al. 2016; Jackson et al. 2016) and the correlation of factors at temporal and spatial scales (Cyronak et al. 2020; Denny and Dowd 2022) in aquatic systems, covariance effects are likely common in nature. Experimental manipulations of positive and negative covariation of temperature and food availability have supported general theoretical expectations showing a significant covariance effect on Daphnia life-history traits (Koussoroplis and Wacker 2016). Negative covariation between factors showed the lowest trait values mirroring the synergistic effects of these factors measured under constant conditions. Interestingly, in the positive covariation scenario the covariance effect on performance was higher than inferred by modeling (Koussoroplis and 23782222, 0, Downloaded from https://silopbus.onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://online.ibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GEOMAR Helmholz Centre of Ocean, Wiley Online Library on [01/1/2022]. See the Terms and Conditions (https://online.ibrary.wiley.com/doi/10.1002/ol.21.0286 by HGF GE

Wacker 2016). As for single factors, gradual phenotypic plasticity can lead to time-dependent effects in multifactorial scenarios and result in deviations of observed trait values from predictions (Koussoroplis et al. 2019). An example in *Daphnia* showed that the covariance effect (effect size and direction) of two fluctuating resource regimes on growth is mediated by fluctuation frequency and, therefore, time for acclimation as well as nutritional reserve effects (Koussoroplis et al. 2019).

It is important to note that beyond the illustrative examples described here, other environmental variables also interact in complex ways altering performance of a variety of biological systems and therefore highlighting the relevance of considering such complex multifactorial scenarios in variability research. Temperature has been shown to interact with irradiance affecting phytoplankton growth rate, and the effect of this interaction is shaped by the photoperiod regime (light fluctuations, Theus et al. 2022). Furthermore, other relevant interactive effects of environmental factors on performance have been highlighted in the literature, like the combined effects of temperature and oxygen concentration on marine invertebrates and fish metabolism and physiology (Pörtner 2010; Vaquer-Sunyer and Duarte 2011; Roman et al. 2019), or temperature and salinity effects on mussel's vital rates (Guzmán-Agüero et al. 2013), but less is known about these interactions under fluctuating conditions.

Challenges of integrating inferences from experimental data and theory on effects of environmental variability

Variety of experimental approaches

Experimental investigations on the effects of environmental fluctuations on biological systems often show contradictory results (i.e., different sizes and direction of variability effects). Marine organisms have shown neutral, positive, and negative acclimation responses to thermal fluctuations (Pansch et al. 2018; Morón Lugo et al. 2020; Vajedsamiei et al. 2021b), which may be attributed to differences among systems (natural variability driving adaptation), type of organisms (short- vs. long-lived), and the measured response (Jackson et al. 2021). The thermal dependence of measured traits (i.e., thermal performance curves) can vary substantially across levels of biological organization, and thus performance depends on the response variable that is measured. For example, thermal tolerance can decline when scaling up from molecular levels to populations as shown in terrestrial ecosystems (Rezende and Bozinovic 2019). This may influence estimates of potential detrimental effects of variability on performance and species persistence at high temperatures. Mismatches can also originate from the experimental scale used to test predictions. For long-lived organisms, short-term thermal performance curves measured during a thermal ramp (e.g., increasing at hourly intervals) might differ from longterm thermal performance curves. This discrepancy can generate differences between observations and expectations under

diel thermal fluctuations because different scales of environmental change are apparent (Rezende et al. 2014). Such aspects are, at least in part, a consequence of the lack of common criteria for including variability components at different scales and biological levels of organization (Thompson et al. 2013; Colinet et al. 2015), which challenges the attempt to generalize across systems and test predictions.

Information about different environmental factors

For environmental variables other than temperature the information is sparse, and approaches differ among variables, but some comparisons between constant and fluctuating environments are available. Variable light affects phytoplankton performance differently than constant light (Shatwell et al. 2012; Theus et al. 2022) and the effect of fluctuations depends on the average irradiance and the period of fluctuations (i.e., the part of the response curve that is analyzed and the frequency of fluctuations) (Litchman 1998, 2000; Shatwell et al. 2012). Experimental studies using salinity gradients showed that marine mussels' vital rates (filtration, respiration, among others) might have nonlinear response curves (Guzmán-Agüero et al. 2013; Peteiro et al. 2018), but the effects of salinity fluctuations have not been addressed using the performance framework. However, an experiment in the estuarine copepod Acartia tonsa, showed that salinity fluctuations decreased gross growth efficiency in comparison to constant environments with the same average salinity (Martínez et al. 2020). Since changes in salinity require osmoregulation by the organisms (Evans and Kültz 2020), such effects might be associated with an increased energetic cost related to adjustments in osmoregulation (Martínez et al. 2020). It has also been shown that coralline microalgae decline growth rate in response to daily pH fluctuations (Cornwall et al. 2013). However, opposite patterns of growth responses to pH fluctuations in corals and coralline algae are shown (summarized in Rivest et al. 2017) as well as for other vital rates (e.g., calcification rates, Cornwall et al. 2018). Overall, knowledge of biotic processes under environmental fluctuations is limited in comparison to constant environments, but also the applied approaches differ among variables (e.g., the performance framework has been to date mostly applied for temperature, Denny and Dowd 2022).

Mismatch between theoretical predictions and experimental results

In addition to experimental approaches, theoretical studies have increasingly addressed the effects of variability on performance, but model predictions often lack experimental tests required to extrapolate their expectations to ecologically relevant conditions (Morash et al. 2018). How accurately reaction norms established for constant or ramp treatments anticipate the performance of ecological systems experiencing environmental fluctuations has been tested in aquatic ecosystems, and many studies showed deviations between observed and predicted patterns for a variety of organisms (anuran larvae, phytoplankton, macroinvertebrates: Niehaus et al. 2012; Wang et al. 2019; Marshall et al. 2021 for temperature, and Litchman 2000; Theus et al. 2022 for light). These mismatches can be instructive to understand the resulting biological process that contributed to their occurrence (see Section How to approach biological consequences of environmental variability). Yet, how predictions can be correctly extrapolated to natural systems remains poorly understood. For example, one terrestrial study showed how predictions from static thermal performance curves matched laboratory observations under fluctuations but not under natural conditions (Khelifa et al. 2019). Here, insects showed faster development when colder temperatures were more frequent (i.e., a higher proportion of cold days), possibly due to an adaptive response to seasonal shifts (Khelifa et al. 2019). This highlights that the extrapolation of conceptual (i.e., proof of concept) experimental results to natural conditions must be done with caution and underscores the need of understanding processes underlying more complex scenarios that more closely approximate natural ecosystems.

Incorporation of abiotic and biotic complexity in variability research

Few studies have assessed more complex abiotic (i.e., environmental components of variability and multiple factors) and biotic (i.e., responses at different levels of biological organization and biological interactions) scenarios. Experimental efforts addressing complex environmental dynamics (abiotic complexity), involving not only fluctuations per se, but changes in fluctuation frequency (Pansch and Hiebenthal 2019; Wang et al. 2019; Kunze et al. 2022), predictability (Fey and Wieczynski 2017; Shama 2017; Rescan et al. 2020; Burton et al. 2020), or considering multiple factors (Koussoroplis and Wacker 2016; Koussoroplis et al. 2019) are scarce and comprise a variety of approaches and types of organism addressed (phytoplankton, zooplankton, mussels, and fish). Covariance effects (i.e., including fluctuations of two factors), for example, have been evaluated connecting models and experiments, but examples are rare and mostly limited to simplified laboratory approaches (Koussoroplis and Wacker 2016; Koussoroplis et al. 2019). Empirical tests across environmental gradients and ecologically relevant scenarios (i.e., representing a wide range of levels and combination of factors that simulate natural variance and covariance effects; e.g., Wahl et al. 2021) are much needed for covering different components of variability and multiple factor effects.

When considering investigations at higher biological levels of organization beyond physiological responses, for example, population and community level responses, two important aspects are added: inter- and intra-specific competition. Specifically, fluctuation-dependent mechanisms such as the storage effect can play a key role in the maintenance of biodiversity when community responses to environmental variability allow for reduced inter-specific competition and stable coexistence of species (Chesson 2000; Descamps-Julien and Gonzalez 2005). These expectations were supported by experiments testing phytoplankton diversity responses to fluctuating vs. constant light (Litchman 1998; Flöder et al. 2002). However, experiments evaluating the effects of environmental variability on phytoplankton communities have shown that species richness can decrease (Burgmer and Hillebrand 2011) or increase (Gerhard et al. 2019) as a consequence of temperature fluctuations. Moreover, it has been proposed that there is a strong link between functional biodiversity and environmental variability (Hodapp et al. 2016; Guislain et al. 2019), suggesting that variability may have drastically different effects on diverse communities compared to the individual or population level. Although there is little experimental work evaluating how diversity mediates the effects of environmental variability on community functions, laboratory experiments showed that communities composed of higher numbers of species can buffer the negative effects of thermal variance on phytoplankton biomass (Bestion et al. 2021) and aquatic fungal leaf decomposition (Gonçalves et al. 2015). Interestingly, studies evaluating the effects of fluctuations on communities generally found that few (likely) well-adapted species (e.g., species with wider thermal range) dominate the communities under fluctuating conditions (Burgmer and Hillebrand 2011; Rasconi et al. 2017; Gerhard et al. 2019; Zhang et al. 2019; Bestion et al. 2021; Cabrerizo et al. 2021). This is surprising considering that natural ecosystems are regularly exposed to variability and are composed of diverse communities. Thus, it is important to note that these experiments are often logistically limited (conducted in closed systems, simulating unrealistic variability dynamics), which influences the capacity to realistically simulate and understand the role of variability in diverse communities (and meta-communities). Sometimes, the effect of variability may also be subtle when looking at species composition, but more pronounced in other measures, such as biochemical composition (Marzetz and Wacker 2021). The presence of such complex ecological processes makes it difficult to design, conduct, and interpret experimental studies and their connection with theory.

Increasing biological complexity typically results in large divergence between model predictions and experimental outcomes. How environmental variability is dampened or amplified in communities of more than one trophic level has been primarily approached by models (Fey and Vasseur 2016; Dee et al. 2020; Simon and Vasseur 2021). Conversely, many experiments (especially large mesocosm experiments) use scenario-based approaches instead of theory-based hypotheses or model assumptions. This was highlighted by Kharouba and Wolkovich (2020) who showed how theory and observations were disconnected in climate change driven phenological mismatch since collected data failed to test for theoretical assumptions (i.e., lack of pairwise per capita fitness) and to define a baseline (defining the range of natural variation in the timing of species interaction before climate change effects occur). The mismatch between theory and empirical work has been recently pointed out, beyond environmental variability research, showing that empirical ecologists often conduct experiments without considering theoretical projections (Korell et al. 2020), while theoretical ecologists use models limited by their assumptions for addressing more realistic scenarios (Klausmeier et al. 2020). This suggests that the lack of a common language, exchange and understanding between theorists and experimentalists might cause mismatches (Grainger et al. 2022). When criticizing that few experimental studies are hypothesis-driven, is that so because there is a lack of theoretical predictions for complex systems that could be used as hypotheses, or do we simply fail to translate existing theoretical predictions for our specific experimental system? Similarly, when arguing that modeling results are often based on unrealistic assumptions, and therefore of little use for predictions, should we not also acknowledge that many experiments are performed at conditions that seem experimentally feasible or promise a strong response to act as a proof-of-concept, thus also compromising on realism?

Dealing with mismatches between theoretical predictions and experimental data

The origin of mismatches

Amending mismatches between (quantitative) models and empirical data is facilitated by first elucidating their origins (see Grainger et al. 2022 for a general synthesis). The first aspect to be considered in variability research is that models are commonly parametrized using experimental data produced under constant conditions in which biological processes that could manifest in variable environments and drive performance cannot be detected. For example, physiological plasticity might produce experimental results not considered in the design of previous models due to local acclimation or adaptation (Sanford and Kelly 2011; Peck et al. 2014). Thus, failing to include such biological processes will lead to mismatches between theoretical predictions on one hand, and experimental results and field observations on the other hand. Yet, even if the relevant biological mechanisms are included in the model, defining the temporal and/or spatial scale at which they operate can be a challenge. Mismatches between predictions and observations can arise, if, for example, the modeled rate of acclimation is too slow or too fast relative to reality (Denny and Dowd 2022). A recent conceptual framework in fluctuating environments could help designing experiments that address the issue of acclimation, the temporal scales at which it operates as well as its consequences on performance (Fey et al. 2021). Such experiments, although logistically challenging for organisms larger than microbes, could help in the design and parametrization of better models.

Another major cause for mismatches to be careful about can be that the model used was originally designed for a different purpose (*see* Klausmeier et al. 2020 for an example of careful model selection and extension). Here, multifactorial variability approaches pose a difficulty due to the interaction between factors. If a model is grounded on experiments for single factors variability, such interactions will not a priori be included in the model and may thus yield mismatches with data from multifactorial experiments (e.g., cross-dependence and covariance effects, see Section Integrating multifactorial variability). Similarly, if a model is specified to describe the physiology of individual organisms, then using the model to infer population level variability effects will ignore interactions within the population that may, for example, give rise to density-dependent effects. This equally holds for models designed for single populations that are used to infer community responses (e.g., if potential trophic cascades of variability effects are not considered, see Hunsicker et al. 2011 for a discussion on scale-dependence of predator-prey interactions). Clearly, mixtures of all the above cases can occur, showing that agreement between model predictions and experimental data are generally challenging to achieve.

Addressing mismatches

For bridging the gaps between variability theory and experimental work, it is important to start with the main questions: What do we, as scientists interested in multidimensional variability effects, expect from theory? And which information can be provided to improve model performance and predictions? Models can be combined with empirical information with different aims: (1) to state null-hypotheses that can be tested by experiments. For example, using nonlinear averaging we can predict performance responses to fluctuating environments in the absence of phenotypic plasticity, and mismatches allow for postulation of such time-dependent effects (e.g., Khelifa et al. 2019; Koussoroplis et al. 2019); (2) to improve mechanistic understanding of experimental data. To study, for example, how physiological processes are affected and interact in variable environments, parameter estimates and an iterative process between experiments and modeling are needed, which is aided by formalized model selection strategies such as AIC (this procedure is important but usually not documented in publications); or (3) to predict or extrapolate species responses to nontested scenarios (e.g., of climate change). For example, having a mechanistic understanding (e.g., physiological thermal limits) improves predictions of potential expansion in alien species (Buckley et al. 2011; Wesselmann et al. 2021) or changes in marine virushost dynamics (Demory et al. 2021). Based on these objectives a variety of models can be used for investigating environmental variability effects on biotic systems including mechanistic and phenomenological types.

Mechanistic models propose relationships between variables such as biomass or abundance that are based on the biological processes and their assumed dependence on environmental factors and variability therein (e.g., dynamic energy budget models, Koussoroplis et al. 2019), or models based on the metabolic theory of ecology (Walters et al. 2012). Phenomenological (or statistical) models aim to best describe data sets and propose relationships between variables and sensitivities on environmental variability based on these empirical or simulated patterns (Schwager et al. 2006; Picoche and Barraquand 2020). There are also other types of models between these extremes, for example, species distribution models that can be correlative or mechanistic and can describe and predict the response of species to environmental variability patterns (Kearney and Porter 2009; Zurell et al. 2009; Dormann et al. 2012; Bocedi et al. 2021), and food web models that are sometimes less mechanistic on the physiological level, but include species interactions and how effects of environmental variability propagate through communities (Raatz et al. 2019; Demory et al. 2021; Quévreux et al. 2021; Simon and Vasseur 2021).

Importantly, successful integration of modeling results and experimental observations is strongly facilitated by early agreement on the conditions that are simulated and parameters that are measured experimentally. An iterative process where model refinements are repeatedly checked against available data is also highly beneficial, pointing out necessary additional experiments to be performed to then feed back into the model formulation. Understanding the scope, strengths, and limitations of approaches that span theory, experiments, and field observations helps this process (Fig. 1; Table S1). Finally, a mismatch between theoretical predictions and experimental data is not necessarily to be viewed as being negative but could instead be used to elucidate where the formulation of our current understanding remains incomplete and foster avenues for further research.

Overcoming experimental challenges related to environmental variability

Relevant aspects for experimental design

To evaluate the feasibility of an experiment and to estimate the number of treatments needed, we must define environmental factors to be manipulated, components of variability, type of organisms (considering life span), level of biological organization, and response traits we aim to address, as well as the spatial and temporal scale, in accordance with our research question. Most experimental setups are restricted by logistics so involving multiple factors and variability easily increases the complexity of the setup. Thus, it is critical to consider aspects that set the number of experimental units when designing such studies:



Fig. 1. When deciding among approaches to investigate variability and its consequences across scales, it is critical to clearly define the research question. All approaches have different advantages and limitations (*see* Table S1 for a detailed description). Generally, moving from controlled to natural systems gains ecological realism while compromising on mechanistic understanding (Petersen et al. 2009; Boyd et al. 2018). Integrating experiments with models requires previous agreement on the conditions that are simulated and parameters that should be measured. Biorender.com was used for creating parts of the figure.

Gerhard et al.

- i. Experimental scale: The scale of an experiment is usually related to the number of experimental units that can be handled (Petersen et al. 2009). Small-scale experiments (e.g., using cell wells or bottles) allow for a larger number of units or replications than large-scale arenas (e.g., mesocosms) that are often limited in the number of units (Fig. 2a). In addition to size (i.e., the volume of experimental units), the complexity of the setup may also limit replication as in the case of highly automated systems (e.g., chemostats to mesocosm infrastructure: Wahl et al. 2016; Pansch and Hiebenthal 2019; Vajedsamiei et al. 2021a). A negative correlation between the experimental scale and the number of units (Fig. 2a) restricts the type of experiment that can be conducted and often enforces scenarios-based over mechanistic approaches (but see Boyd et al. 2018 for alternative hybrid approaches such as collapsed and main vector designs). It is also important to consider that spatial and temporal scales in experiments are determined by the biological level of organization (e.g., individuals, communities), type of response (short- or long-term), and type of organisms (life span, generation time) that are investigated (Ryo et al. 2019; Jackson et al. 2021; Denny and Dowd 2022). Depending on its generation time and the frequency of fluctuations, the same environmental variability can be experienced as environmental fluctuations in long-lived organisms or as press disturbances in short-lived organisms, for which the fluctuation frequency is higher than the generation time (Jackson et al. 2021). Since the experience of past environmental conditions might affect biological responses to current conditions (a phenomenon defined as ecological memory, Jackson et al. 2021), responses to variability might differ substantially across types of organisms and traits measured. While longlived organisms respond to fluctuations according to previous exposure to variability (acclimation), short-lived organisms undergo continuous generations that experience different "static" environments where the parental environmental information differs from the experienced by the offspring (parental effect/epigenetic plasticity; Jackson et al. 2021; Kunze et al. 2022). Thus, it is fundamental to consider biological and environmental variability scales in experimental designs and elaborate consistent hypotheses.
- ii. *Theoretical approaches*: Using theoretical approaches to define the expected type of responses is important for elaborating hypotheses, either when adopting a general framework or when testing model predictions (Grainger et al. 2022) as well as to establish the treatments (factor selection, levels of factors, components of variability). More specifically, theoretical information can be also used to define the minimum number of treatment levels necessary per factor (Fig. 2b). For example, linear and nonlinear responses need at least two and three points to be detected, respectively. If the shape of the response to the target factors is unknown, previous small-scale experimental trials should be developed informing the design of the main experiment (Morel-Journel et al. 2020).

- iii. Components of variability: Choosing which component(s) of variability are experimentally manipulated and the number of levels of each treatment depends on the aim (e.g., testing a future environmental scenario or isolating the impact of one component of variability). Each component of variability (variance, frequency, and autocorrelation; Fig. 2c) requires different manipulations of the environmental variable of interest and can be manipulated simultaneously or independently, though the manipulation of one may necessarily affect another (e.g., total variance and frequency can be manipulated independently, but frequency manipulations affect autocorrelation and rates of change). Experiments might include the manipulation of different components of variability in a gradient design to evaluate potential threshold or nonlinear biological responses, but the inclusion of more than one component of variability increases the complexity of the setup. Factorial designs might be simpler for combining components of variability, but less informative about the biological responses to the combination of components. In these cases, only one environmental factor is considered (e.g., temperature) in the experimental design, but multiple explanatory variables are included (e.g., temperature variance and frequency of fluctuations, and therefore its combined effects).
- iv. Multifactorial scenarios: The number of environmental factors (e.g., light, nutrients, temperature), as well as their identity and levels, determine potential nonadditive effects. Such effects can be complex when covariance (Koussoroplis and Wacker 2016; Koussoroplis et al. 2019) or interactive affect across environmental gradients (Koussoroplis et al. 2017) are present since the direction and magnitude of effects may change and/or vary at different levels of the investigated factors. This type of nonadditive effects increases the number of treatments needed for their investigation, since gradient designs and response surfaces (full factorial designs) are needed for evaluating the effects at different levels (Fig. 2d). Including multiple environmental factors implies complex experimental designs where decisions must be made about what factors, levels, and components of variability are included to capture the relevant regions of biological responses.

Relevant aspects for variability manipulations

There are key aspects that challenge our capacity to investigate and understand the role of variability through experimental research. Many of these aspects are related to experimental manipulations and are shared with any other experimental approach (Campbell and Stanley 1963; Petersen et al. 2009; Boyd et al. 2018), but here those are considered from a variability perspective.

The initial conditions used for experimental setups might have important effects on the responses measured depending on the organism or community type and origin. Defining



Fig. 2. Aspects to be considered for experimental designs and analyses involving environmental variability and multiple factors. (**a**) Negative relationship (displayed on log–log scale) of the number of experimental units that are usually handled and the maximum volume of these units (as an estimation of the scale) ranging from large-scale enclosure experiments to small microcosm experiments (microcosms = correspond to other type of microcosm). Data were obtained from an online survey (Appendix S2). (**b**) Different types of response shape (e.g., linear and nonlinear) to environmental gradients can be expected and determine the minimum number of treatments needed. (**c**) Different components of variability can be investigated and combined when considering temporal or spatial variability conditioning the number of treatments needed. (**d**) Considering multiple factors, nonadditive effects can be investigated using different approaches that differ in the treatment demand. Nonadditive effects can be analyzed by factorial designs where the effect size of two or more factors are tested independently and together identifying the type of effect, or by using response surfaces, which allow for testing relevant areas (combination of factor levels) with high effect sizes. The effects of variability might depend on the type of interaction among factors for the considered levels. Two factors can, for example, show synchrony in their dynamics (in-phase, positive covariation) or diverge over time (out-of-phase, negative covariation) having important consequences for organisms in changing environments (e.g., positive covariation used as cue for anticipatory responses to change, or negative covariation might lead to detrimental effects). Furthermore, changes in one factor might affect the response to a second factor causing cross-dependence and a different expected impact of its variability.

aspects as pre-experimental acclimation, control treatments, and how to manipulate environmental variability are therefore crucial, but far from trivial. Manipulating different components of variability (e.g., reducing or increasing the total variance) are logistically not easy, especially in field experiments (e.g., outdoor in situ mesocosms placed in

Environmental variability in aquatic ecosystems

aquatic systems or in the field, Fig.1) where nonmanipulated environmental variation may generate idiosyncratic effects (i.e., noise) or restrict the simulation of some components of variability (e.g., the manipulation of thermal variance without changing the mean) (Hong and Shurin 2015; Fey and Wieczynski 2017). Such limitations might lead to the use of extreme variability treatments that extend beyond ecological relevance (Korell et al. 2020) or to the underrepresentation of natural variability (Ziegler et al. 2021), imposing additional bias on our understanding. Thus, the integration of highly and less controlled experiments is needed to approach different aspects (Fig. 1).

When addressing the role of environmental variability in a global change context, we can discriminate between three approaches: (1) the acceptance of natural (deterministic and stochastic) variability. This approach allows experiments to be more natural (and realistic) and is a valuable approach when testing the effects of changing mean conditions, but variability components are assumed constant over time (Fig. 3, e.g., Wahl et al. 2016; Yvon-Durocher et al. 2017; Barneche et al. 2021; Sawall et al. 2021), (2) the manipulation of variability per se (compared to constant regimes) or the manipulation of particular components of variability under highly controlled conditions where nonmanipulated factors remain constant. To explicitly test for the role of variability and its components, mechanistic approaches may be applied (Fig. 3, Colinet et al. 2015), and (3) the combination of both approaches (i.e., accepting particular aspects of variability, while manipulating others). This may be an important step in understanding the role of complex variability patterns in aquatic ecosystems (Fig. 3).

Extracting information from local environmental time series for characterizing means and variability components allows for setting appropriate experimental treatments (Pansch et al. 2018; Cabrerizo et al. 2021; Dobry et al. 2021; Kroeker et al. 2021; Wolf et al. 2022). For multiple-driver experiments, characterizing the main drivers in a system and how they covary (e.g., Wahl et al. 2021) is vital for designing ecologically relevant manipulative experiments with a feasible number of treatments (i.e., reducing the complexity of the setting by selecting the combination of factors that show high variation or covariation in nature and combine them with future scenarios). Variability regimes identified like this can be manipulated under highly controlled experiments (i.e., laboratory conditions simulating natural regimes) or on top of natural environmental variability (Fig. 3). The use of such approaches is commonly related to the type of experiment (laboratory, outdoor/indoor mesocosms, etc.), and its characteristics (Fig. 1; Table S1). Hence, defining the control in such different approaches is key and may range from complete elimination of variability, to reduced or manipulated variability, to the natural in situ variability, depending on the research question and hypothesis (Fig. 3).

In addition to conducting experiments at different scales, the comparison of experiments at the same scale but with different degrees of controlled variability (e.g., indoor tanks vs. outdoor enclosures) may allow to disentangle noise from the manipulated variability. Large coordinated mesocosm experiments across sites (Landkildehus et al. 2014; Mahdy et al. 2015), latitudes (Lenz et al. 2011, 2018), and repeated over time (Nejstgaard et al. 2006; Larsen et al. 2015) can also facilitate the generalization and therefore synthesis of patterns by ruling out idiosyncratic (noise) effects, and by including natural variability of nonmanipulated factors relevant for the investigated sites (Urrutia-Cordero et al. 2021). Such spatial and temporal extension of experiments is an important aspect since the majority of ecological studies are traditionally carried out in northern temperate systems (Martin et al. 2012; Thomsen et al. 2014), but diurnal and annual natural variability ranges differ among climatic regions and hemispheres (Wang and Dillon 2014). However, it is relevant to consider that these efforts need thoughtful planning and designing to generate comparable results; including aspects like the characteristics of the facilities (experimental units might not be identical), and protocols for sampling and analysis (Fraser et al. 2013; but see e.g., Lenz et al. 2011, 2018 for a globally comparable approach).

Connecting patterns to processes

Experimental approaches vary in the capacity to investigate mechanisms underlying observed patterns (Fig.1; Table S1; Boyd et al. 2018). In the experimental manipulation of environmental variability (variance, frequency, or predictability) that is imposed across treatments, statistical tests are primarily designed to determine whether differences exist between or among treatments, and whether these differences change across time. This framing poses two related issues. First, albeit rarely addressed, when differences were not observed this might be because: (1) the amount of environmental variation present was not sufficient to produce a measurable biological or ecological response; (2) the response variables measured are not relevant for the environmental change simulated (i.e., do not reflect how organisms respond to the manipulated environmental aspect); or (3) different environments (e.g., different types of environmental variation) produced impacts on biological processes that coincidently yield similar values for a given response variable. Fey et al. (2021) resolved environmental domains where phytoplankton species that acclimate rapidly vs. gradually can exhibit the same overall growth rates in certain variable environments, despite having different biological responses to this variation. As such, studies should explore the potential for cryptic biological processes (Strauss 2014) when no treatment differences exist. Second, when treatments differ from others (or from null models), it can be instructive to understand the underlying biological process that contributed to this occurrence.

For expanding our understanding of the processes driving observed patterns, mechanistic and scenario-based experiments can be integrated in a synergistic way. Combining experimental scales increases the number of treatments that



Fig. 3. Different experimental approaches can be used for manipulating environmental variability, but there is a lack of methodological consistency in this research area. Here we conceptualize three general approaches and provide examples for them: (a) Investigating components of variability under controlled conditions: This approach aims for testing specific components of variability (e.g., mean, variance, extremes) and their interactions, maintaining other aspects constant. This approach is mostly used in controlled laboratory experiments but a variety of methods for simulating variability have been implemented, challenging comparisons among studies (e.g., variance simulated by gradual vs. rapid changes; *see* Colinet et al. 2015 for a detailed discussion). (b) Using natural environmental variability as background conditions. In this case, natural variability is maintained for some factors or components of their variability while others are modified. For example, natural variance can be kept while a change in mean is applied or daily and stochastic variability. This approach is mostly used in mesocosm experimental units. Alternatively, extremes such as heatwaves can be simulated on top of natural variability. This approach is mostly used in mesocosm experiments (*see* Thompson et al. 2013 for a detailed discussion). (c) Translating natural variability measured in the field into experimental designs. Different variability components of nor more factors are manipulated according to natural levels and dynamics (e.g., covariation, order of events, rates of changes), and can be compared to future-predicted scenarios or stress events. This approach can be implemented in controlled laboratory experiments or by using natural environmental variability as background depending on the manipulated factor.

can be tested and allows for addressing different levels of organization (i.e., individual, population, community, ecosystem) by increasing the number of units and integrating experimental scales (Figs. 1, 2). We here suggest three ways of combining experimental approaches: i. *Upscaling in complexity*: Patterns and mechanisms explaining variability effects shown in simplified models and laboratory experiments can be scaled to more biologically complex scenarios. In this approach it is key to conduct hypothesis-based experiments, and to measure

parameters that help improve models in an iterative process. One way is to identify ecologically relevant scales of variation by measuring natural environmental variability across temporal and spatial scales of multiple drivers, and to test realistic combinations of such factors in the laboratory to detect relevant scales of biological responses at low levels of complexity. In a second step, additional complexity (e.g., additional trophic levels) should be incorporated in mesocosms (e.g., Wahl et al. 2021). Testing the direct effects of variability on primary producers and consumers separately and together may disentangle how indirect effects are translated into the next trophic level and interact with direct effects. When upscaling experimental evaluations from simple (e.g., artificial laboratory) to more complex (e.g., mesocosm or field) studies it is important to consider that potential biological functions, such as development of antipredator responses, that might not be present under simplified environmental conditions but are important in more complex systems (Nejstgaard et al. 2007).

- ii. Downscaling in complexity: This approach can be used for discerning mechanisms underlying field observations and explain mechanisms in long-term variability patterns. Imposing ecologically relevant patterns of variability (obtained through field measurements) in mesocosm experiments, and collecting high-resolution data on species or community traits and in situ processes may give clues regarding the underlying mechanisms driving patterns that remain cryptic would otherwise (Yvon-Durocher et al. 2010; Wahl et al. 2021). Outcomes can be used for generating hypotheses to be explicitly tested in controlled laboratory experiments that follow the mesocosm experiments (Yvon-Durocher et al. 2017).
- iii. *Parallel-complementary experiments*: Highly controlled and replicated laboratory experiments can be conducted in parallel to large-scale mesocosm efforts to identify response curves to an environmental gradient, or to extend variability treatments to facilitate the interpretation of patterns observed in the accompanying mesocosm experiment. Vajedsamiei et al. (2021*c*), for example, performed repeated side incubations of marine mussels originating from large-scale and long-term mesocosm experiments, which helped to identify the physiological mechanisms driving the long-term outcomes. Such complementary experiments or as bottles or enclosures placed in the mesocosms, depending on the manipulated factor.

Conclusions

Aquatic ecology has advanced in incorporating environmental variability across multiple environmental factors in recent theoretical and empirical studies. However, the complexity that this research area involves makes it crucial to use a common framework and have consensus on definitions that strengthen the integration of generated knowledge and the design of comparable experimental studies for dealing with variability. A deeper integration of disciplines and development of broader realistic approaches is needed for understanding and predicting present and future global change. Suited experimental approaches may stem from experiments driven by theory-derived hypotheses, scaling predictions and combining different types of experimental approaches.

References

- Angilletta, M. J., Jr. 2006. Estimating and comparing thermal performance curves. J. Therm. Biol. **31**: 541–545. doi:10. 1016/j.jtherbio.2006.06.002
- Aranguren-Gassis, M., C. T. Kremer, C. A. Klausmeier, and E. Litchman. 2019. Nitrogen limitation inhibits marine diatom adaptation to high temperatures. Ecol. Lett. 22: 1860– 1869. doi:10.1111/ele.13378
- Barneche, D. R., C. J. Hulatt, M. Dossena, D. Padfield, G. Woodward, M. Trimmer, and G. Yvon-Durocher. 2021. Warming impairs trophic transfer efficiency in a long-term field experiment. Nature **592**: 76–79. doi:10.1038/s41586-021-03352-2
- Berger, S. A., S. Diehl, H. Stibor, P. Sebastian, and A. Scherz. 2014. Separating effects of climatic drivers and biotic feedbacks on seasonal plankton dynamics: No sign of trophic mismatch. Freshw. Biol. **59**: 2204–2220. doi:10.1111/fwb. 12424
- Bernhardt, J. R., J. M. Sunday, P. L. Thompson, and M. I. O'Connor. 2018. Nonlinear averaging of thermal experience predicts population growth rates in a thermally variable environment. Proc. R. Soc. B: Biol. Sci. 285: 20181076. doi:10.1098/rspb.2018.1076
- Bestion, E., C.-E. Schaum, and G. Yvon-Durocher. 2018. Nutrient limitation constrains thermal tolerance in freshwater phytoplankton. Limnol. Oceanogr.: Lett. **3**: 436–443. doi: 10.1002/lol2.10096
- Bestion, E., B. Haegeman, S. Alvarez Codesal, A. Garreau, M. Huet, S. Barton, and J. M. Montoya. 2021. Phytoplankton biodiversity is more important for ecosystem functioning in highly variable thermal environments. Proc. Natl. Acad. Sci. USA **118**: e2019591118. doi:10.1073/pnas.2019591118
- Bocedi, G., S. C. Palmer, A.-K. Malchow, D. Zurell, K. Watts, and J. M. Travis. 2021. RangeShifter 2.0: An extended and enhanced platform for modelling spatial eco-evolutionary dynamics and species' responses to environmental changes. Ecography **44**: 1453–1462. doi:10.1111/ecog. 05687
- Boyd, P. W., C. E. Cornwall, A. Davison, S. C. Doney, M. Fourquez, C. L. Hurd, I. D. Lima, and A. McMinn. 2016. Biological responses to environmental heterogeneity under

future ocean conditions. Glob. Change Biol. **22**: 2633–2650. doi:10.1111/gcb.13287

- Boyd, P. W., and others. 2018. Experimental strategies to assess the biological ramifications of multiple drivers of global ocean change—A review. Glob. Change Biol. **24**: 2239–2261. doi:10.1111/gcb.14102
- Buckley, L. B., S. A. Waaser, H. J. MacLean, and R. Fox. 2011. Does including physiology improve species distribution model predictions of responses to recent climate change? Ecology **92**: 2214–2221. doi:10.1890/11-0066.1
- Burgmer, T., and H. Hillebrand. 2011. Temperature mean and variance alter phytoplankton biomass and biodiversity in a long-term microcosm experiment. Oikos **120**: 922–933. doi:10.1111/j.1600-0706.2010.19301.x
- Burton, T., H. Lakka, and S. Einum. 2020. Measuring phenotypes in fluctuating environments. Funct. Ecol. **34**: 606– 615. doi:10.1111/1365-2435.13501
- Cabrerizo, M. J., E. Marañón, C. Fernández-González, A. Alonso-Núñez, H. Larsson, and M. Aranguren-Gassis. 2021.
 Temperature fluctuation attenuates the effects of warming in estuarine microbial plankton communities. Front. Mar. Sci. 8: 656282. doi:10.3389/fmars.2021.656282
- Chesson, P. 2000. General theory of competitive coexistence in spatially-varying environments. Theor. Popul. Biol. **58**: 211–237. doi:10.1006/tpbi.2000.1486
- Chesson, P., M. J. Donahue, B. A. Melbourne, and A. L. Sears. 2005. Scale transition theory for understanding mechanisms in metacommunities, p. 279–306. *In* M. Holyoak, M. A. Leibold, and R. D. Holt [eds.], *Metacommunities: spatial dynamics and ecological communities*. Univ. Chicago Press.
- Coble, A. A., R. G. Asch, S. Rivero-Calle, S. M. Heerhartz, J. M. Holding, C. T. Kremer, M. Finiguerra, and K. E. Strock. 2016. Climate is variable, but is our science? Limnol. Oceanogr.: Bull. 25: 71–76. doi:10.1002/lob.10115
- Colinet, H., B. J. Sinclair, P. Vernon, and D. Renault. 2015. Insects in fluctuating thermal environments. Annu. Rev. Entomol. **60**: 123–140. doi:10.1146/annurev-ento-010814-021017
- Colinet, H., J. P. Rinehart, G. D. Yocum, and K. J. Greenlee. 2018. Mechanisms underpinning the beneficial effects of fluctuating thermal regimes in insect cold tolerance. J. Exp. Biol. **221**: jeb164806. doi:10.1242/jeb.164806
- Cornwall, C. E., C. D. Hepburn, C. M. McGraw, K. I. Currie, C. A. Pilditch, K. A. Hunter, P. W. Boyd, and C. L. Hurd. 2013. Diurnal fluctuations in seawater pH influence the response of a calcifying macroalga to ocean acidification. Proc. R. Soc. B: Biol. Sci. 280: 20132201. doi:10.1098/rspb. 2013.2201
- Campbell, D. T., and J. C. Stanley. 1963. Experimental and quasi-experimental designs for research, p. 1–71. In N. L. Gage (Ed.), *Handbook of Research on Teaching*. Rand McNally & Company.

- Cornwall, C. E., S. Comeau, T. M. DeCarlo, B. Moore, Q. D'alexis, and M. T. McCulloch. 2018. Resistance of corals and coralline algae to ocean acidification: Physiological control of calcification under natural pH variability. Proc. R. Soc. B: Biol. Sci. 285: 20181168. doi:10.1098/rspb.2018. 1168
- Cyronak, T., and others. 2020. Diel temperature and pH variability scale with depth across diverse coral reef habitats. Limnol. Oceanogr.: Lett. **5**: 193–203. doi:10.1002/lol2. 10129
- De Senerpont Domis, L. N., D. B. Van de Waal, N. R. Helmsing, E. Van Donk, and W. M. Mooij. 2014. Community stoichiometry in a changing world: Combined effects of warming and eutrophication on phytoplankton dynamics. Ecology **95**: 1485–1495. doi:10.1890/13-1251.1
- Dee, L. E., D. Okamtoto, A. Gårdmark, J. M. Montoya, and S. J. Miller. 2020. Temperature variability alters the stability and thresholds for collapse of interacting species. Philos. Trans. R. Soc. Lond. B Biol. Sci. **375**: 20190457. doi:10. 1098/rstb.2019.0457
- Demory, D., J. S. Weitz, A.-C. Baudoux, S. Touzeau, N. Simon, S. Rabouille, A. Sciandra, and O. Bernard. 2021. A thermal trade-off between viral production and degradation drives virus-phytoplankton population dynamics. Ecol. Lett. **24**: 1133–1144. doi:10.1111/ele.13722
- Denny, M. 2016. Ecological variation and its consequences, p. 364–378. *In, Ecological mechanics: Principles of life's physical interactions*. Princeton Univ. Press.
- Denny, M. 2017. The fallacy of the average: On the ubiquity, utility and continuing novelty of Jensen's inequality. J. Exp. Biol. **220**: 139–146. doi:10.1242/jeb.140368
- Denny, M. 2019. Performance in a variable world: Using Jensen's inequality to scale up from individuals to populations. Conserv. Physiol. 7: coz053. doi:10.1093/ conphys/coz053
- Denny, M. W., and W. W. Dowd. 2022. Physiological consequences of oceanic environmental variation: Life from a pelagic organism's perspective. Ann. Rev. Mar. Sci. 14: 25– 48. doi:10.1146/annurev-marine-040221-115454
- Descamps-Julien, B., and A. Gonzalez. 2005. Stable coexistence in a fluctuating environment: An experimental demonstration. Ecology 86: 2815–2824. doi:10.1890/04-1700
- Dillon, M. E., H. A. Woods, G. Wang, S. B. Fey, D. A. Vasseur, R. S. Telemeco, K. Marshall, and S. Pincebourde. 2016. Life in the frequency domain: The biological impacts of changes in climate variability at multiple time scales. Integr. Comp. Biol. 56: 14–30. doi:10.1093/icb/icw024
- Dobry, E., G. Schoeniger, and S. A. Nutile. 2021. The effect of salinity fluctuation in freshwater streams on the fecundity of post-diapause *Chironomus dilutus*. Ecotoxicology **30**: 224–230. doi:10.1007/s10646-020-02338-6
- Dormann, C. F., and others. 2012. Correlation and process in species distribution models: Bridging a dichotomy.

J. Biogeogr. **39**: 2119–2131. doi:10.1111/j.1365-2699.2011. 02659.x

- Dowd, W. W., F. A. King, and M. W. Denny. 2015. Thermal variation, thermal extremes and the physiological performance of individuals. J. Exp. Biol. 218: 1956–1967. doi:10. 1242/jeb.114926
- Evans, T. G., and D. Kültz. 2020. The cellular stress response in fish exposed to salinity fluctuations. J. Exp. Zool. A Ecol. Integr. Physiol. **333**: 421–435. doi:10.1002/jez.2350
- Fey, S. B., and D. A. Vasseur. 2016. Thermal variability alters the impact of climate warming on consumer–resource systems. Ecology 97: 1690–1699. doi:10.1890/15-1838.1
- Fey, S. B., and D. J. Wieczynski. 2017. The temporal structure of the environment may influence range expansions during climate warming. Glob. Change Biol. 23: 635–645. doi: 10.1111/gcb.13468
- Fey, S. B., C. T. Kremer, T. J. Layden, and D. A. Vasseur. 2021. Resolving the consequences of gradual phenotypic plasticity for populations in variable environments. Ecol. Monogr. 91: e01478. doi:10.1002/ecm.1478
- Flanagan, K. M., and E. McCauley. 2010. Experimental warming increases CO_2 saturation in a shallow prairie pond. Aquat. Ecol. **44**: 749–759. doi:10.1007/s10452-010-9313-0
- Flöder, S., J. Urabe, and Z. Kawabata. 2002. The influence of fluctuating light intensities on species composition and diversity of natural phytoplankton communities. Oecologia 133: 395–401. doi:10.1007/s00442-002-1048-8
- Fraser, L. H., and others. 2013. Coordinated distributed experiments: An emerging tool for testing global hypotheses in ecology and environmental science. Front. Ecol. Environ. 11: 147–155. doi:10.1890/110279
- Fujiwara, M., and T. Takada. 2017. Environmental stochasticity, p. 1–8. *In*, *Els*. John Wiley & Sons. doi:10. 1002/9780470015902.a0021220.pub2
- Gerhard, M., A. M. Koussoroplis, H. Hillebrand, and M. Striebel. 2019. Phytoplankton community responses to temperature fluctuations under different nutrient concentrations and stoichiometry. Ecology **100**: e02834. doi:10. 1002/ecy.2834
- Gonçalves, A. L., M. A. Graça, and C. Canhoto. 2015. Is diversity a buffer against environmental temperature fluctuations?—A decomposition experiment with aquatic fungi. Fungal Ecol. **17**: 96–102. doi:10.1016/j.funeco.2015.05.013
- González-Olalla, J. M., J. M. Medina-Sánchez, and P. Carrillo. 2022. Fluctuation at high temperature combined with nutrients alters the thermal dependence of phytoplankton. Microb. Ecol. 83: 555–567. doi:10.1007/s00248-021-01787-8
- Grainger, T. N., and others. 2022. An empiricist's guide to using ecological theory. Am. Nat. **199**: 1–20. doi:10.1086/ 717206
- Guislain, A., B. E. Beisner, and J. Köhler. 2019. Variation in species light acquisition traits under fluctuating light

regimes: Implications for non-equilibrium coexistence. Oikos **128**: 716–728. doi:10.1111/oik.05297

- Gunderson, A. R., E. J. Armstrong, and J. H. Stillman. 2016. Multiple stressors in a changing world: The need for an improved perspective on physiological responses to the dynamic marine environment. Ann. Rev. Mar. Sci. 8: 357– 378. doi:10.1146/annurev-marine-122414-033953
- Guzmán-Agüero, J. E., M. Nieves-Soto, M. Á. Hurtado, P. Piña-Valdez, and M. C. Garza-Aguirre. 2013. Feeding physiology and scope for growth of the oyster *Crassostrea corteziensis* (Hertlein, 1951) acclimated to different conditions of temperature and salinity. Aquacult. Int. **21**: 283–297. doi:10. 1007/s10499-012-9550-4
- Harris, R. M., and others. 2018. Biological responses to the press and pulse of climate trends and extreme events. Nat. Clim. Change **8**: 579–587. doi:10.1038/s41558-018-0187-9
- Hastings, A., and H. Caswell. 1979. Role of environmental variability in the evolution of life history strategies. Proc. Natl. Acad. Sci. USA **76**: 4700–4703. doi:10.1073/pnas.76.9.4700
- Hennemann, M. C., and M. M. Petrucio. 2010. Seasonal phytoplankton response to increased temperature and phosphorus inputs in a freshwater coastal lagoon, southern Brazil: A microcosm bioassay. Acta Theriol. **22**: 295–305. doi:10.4322/actalb.02203006
- Hintz, N. H., B. Schulze, A. Wacker, and M. Striebel. 2022. Ecological impacts of photosynthetic light harvesting in changing aquatic environments: A systematic literature map. Ecol. Evol. **12**: e8753. doi:10.1002/ece3.8753
- Hodapp, D., H. Hillebrand, B. Blasius, and A. B. Ryabov. 2016.
 Environmental and trait variability constrain community structure and the biodiversity-productivity relationship.
 Ecology 97: 1463–1474. doi:10.1890/15-0730.1
- Hong, B. C., and J. B. Shurin. 2015. Latitudinal variation in the response of tidepool copepods to mean and daily range in temperature. Ecology 96: 2348–2359. doi:10.1890/14-1695.1
- Hunsicker, M. E., and others. 2011. Functional responses and scaling in predator–prey interactions of marine fishes: Contemporary issues and emerging concepts. Ecol. Lett. **14**: 1288–1299. doi:10.1111/j.1461-0248.2011.01696.x
- Jackson, M. C., C. J. Loewen, R. D. Vinebrooke, and C. T. Chimimba. 2016. Net effects of multiple stressors in freshwater ecosystems: A meta-analysis. Glob. Change Biol. 22: 180–189. doi:10.1111/gcb.13028
- Jackson, M. C., S. Pawar, and G. Woodward. 2021. The temporal dynamics of multiple stressor effects: From individuals to ecosystems. Trends Ecol. Evol. **36**: 402–410. doi:10. 1016/j.tree.2021.01.005
- Jensen, J. L. W. V. 1906. Sur les fonctions convexes et les inégalités entre les valeurs moyennes. Acta Math. **30**: 175–193. doi:10.1007/BF02418571
- Kapsenberg, L., and T. Cyronak. 2019. Ocean acidification refugia in variable environments. Glob. Change Biol. 25: 3201–3214. doi:10.1111/gcb.14730

- Kearney, M., and W. Porter. 2009. Mechanistic niche modelling: Combining physiological and spatial data to predict species' ranges. Ecol. Lett. **12**: 334–350. doi:10.1111/j. 1461-0248.2008.01277.x
- Kharouba, H. M., and E. M. Wolkovich. 2020. Disconnects between ecological theory and data in phenological mismatch research. Nat. Clim. Change **10**: 406–415. doi:10. 1038/s41558-020-0752-x
- Khelifa, R., W. U. Blanckenhorn, J. Roy, P. T. Rohner, and H. Mahdjoub. 2019. Usefulness and limitations of thermal performance curves in predicting ectotherm development under climatic variability. J. Anim. Ecol. 88: 1901–1912. doi:10.1111/1365-2656.13077
- Kingsolver, J. G., J. K. Higgins, and K. E. Augustine. 2015. Fluctuating temperatures and ectotherm growth: Distinguishing non-linear and time-dependent effects. J. Exp. Biol. **218**: 2218–2225. doi:10.1242/jeb.120733
- Kingsolver, J. G., and H. A. Woods. 2016. Beyond thermal performance curves: Modeling time-dependent effects of thermal stress on ectotherm growth rates. Am. Nat. 187: 283– 294. doi:10.1086/684786
- Klausmeier, C. A., M. M. Osmond, C. T. Kremer, and E. Litchman. 2020. Ecological limits to evolutionary rescue. Philos. Trans. R. Soc. Lond. B Biol. Sci. **375**: 20190453. doi: 10.1098/rstb.2019.0453
- Korell, L., H. Auge, J. M. Chase, S. Harpole, and T. M. Knight. 2020. We need more realistic climate change experiments for understanding ecosystems of the future. Glob. Change Biol. 26: 325–327. doi:10.1111/gcb.14797
- Koussoroplis, A.-M., and A. Wacker. 2016. Covariance modulates the effect of joint temperature and food variance on ectotherm life-history traits. Ecol. Lett. **19**: 143–152. doi: 10.1111/ele.12546
- Koussoroplis, A.-M., S. Pincebourde, and A. Wacker. 2017. Understanding and predicting physiological performance of organisms in fluctuating and multifactorial environments. Ecol. Monogr. 87: 178–197. doi:10.1002/ecm.1247
- Koussoroplis, A., S. Schälicke, M. Raatz, M. Bach, and A. Wacker. 2019. Feeding in the frequency domain: Coarsergrained environments increase consumer sensitivity to resource variability, covariance and phase. Ecol. Lett. **22**: 1104–1114. doi:10.1111/ele.13267
- Kremer, C. T., S. B. Fey, A. A. Arellano, and D. A. Vasseur. 2018. Gradual plasticity alters population dynamics in variable environments: Thermal acclimation in the green alga *Chlamydomonas reinhartdii*. Proc. R. Soc. B: Biol. Sci. **285**: 20171942. doi:10.1098/rspb.2017.1942
- Kroeker, K. J., L. E. Bell, E. M. Donham, U. Hoshijima, S. Lummis, J. A. Toy, and E. Willis-Norton. 2020. Ecological change in dynamic environments: Accounting for temporal environmental variability in studies of ocean change biology. Glob. Change Biol. 26: 54–67. doi:10.1111/gcb. 14868

- Kroeker, K. J., C. Powell, and E. M. Donham. 2021. Windows of vulnerability: Seasonal mismatches in exposure and resource identity determine ocean acidification's effect on a primary consumer at high latitude. Glob. Change Biol. 27: 1042–1051. doi:10.1111/gcb.15449
- Kunze, C., M. Gerhard, M. Jacob, N. A. Franke, M. Schröder, and M. Striebel. 2022. Phytoplankton community performance depends on the frequency of temperature fluctuations. Front. Mar. Sci. 8: 812902. doi:10.3389/fmars.2021. 812902
- Landkildehus, F., and others. 2014. Climate change effects on shallow lakes: Design and preliminary results of a cross-European climate gradient mesocosm experiment. Estonian J. Ecol. **63**: 71–89. doi:10.3176/eco.2014.2.02
- Larsen, A., J. K. Egge, J. C. Nejstgaard, I. Di Capua, R. Thyrhaug, G. Bratbak, and T. F. Thingstad. 2015. Contrasting response to nutrient manipulation in Arctic mesocosms are reproduced by a minimum microbial food web model. Limnol. Oceanogr. 60: 360–374. doi:10.1002/lno.10025
- Lenz, M., and others. 2011. Non-native marine invertebrates are more tolerant towards environmental stress than taxonomically related native species: Results from a globally replicated study. Environ. Res. **111**: 943–952. doi:10.1016/ j.envres.2011.05.001
- Lenz, M., and others. 2018. Heat challenges can enhance population tolerance to thermal stress in mussels: A potential mechanism by which ship transport can increase species invasiveness. Biol. Invasions **20**: 3107–3122. doi:10.1007/ s10530-018-1762-8
- Litchman, E. 1998. Population and community responses of phytoplankton to fluctuating light. Oecologia **117**: 247–257. doi:10.1007/s004420050655
- Litchman, E. 2000. Growth rates of phytoplankton under fluctuating light. Freshw. Biol. 44: 223–235. doi:10.1046/j. 1365-2427.2000.00559.x
- Litchman, E., P. de Tezanos Pinto, K. F. Edwards, C. A. Klausmeier, C. T. Kremer, and M. K. Thomas. 2015. Global biogeochemical impacts of phytoplankton: A trait-based perspective. J. Ecol. **103**: 1384–1396. doi:10.1111/1365-2745.12438
- Mahdy, A., and others. 2015. Effects of water temperature on summer periphyton biomass in shallow lakes: A pan-European mesocosm experiment. Aquat. Sci. **77**: 499– 510. doi:10.1007/s00027-015-0394-7
- Marshall, K. E., and others. 2021. Whole-organism responses to constant temperatures do not predict responses to variable temperatures in the ecosystem engineer *Mytilus trossulus*. Proc. R. Soc. B: Biol. Sci. **288**: 20202968. doi:10. 1098/rspb.2020.2968
- Martin, L. J., B. Blossey, and E. Ellis. 2012. Mapping where ecologists work: Biases in the global distribution of terrestrial ecological observations. Front. Ecol. Environ. **10**: 195–201. doi:10.1890/110154

- Martínez, M., L. Rodríguez-Graña, L. Santos, A. Denicola, and D. Calliari. 2020. Long-term exposure to salinity variations induces protein carbonylation in the copepod *Acartia tonsa*. J. Exp. Mar. Biol. Ecol. **526**: 151337. doi:10.1016/j.jembe. 2020.151337
- Marzetz, V., and A. Wacker. 2021. Evaluating the relevance of species sorting and physiological plasticity of phytoplankton communities grown in a multifactor environment. Freshw. Biol. **66**: 1992–2003. doi:10.1111/fwb.13810
- Morash, A. J., C. Neufeld, T. J. MacCormack, and S. Currie. 2018. The importance of incorporating natural thermal variation when evaluating physiological performance in wild species. J. Exp. Biol. **221**: jeb164673. doi:10.1242/jeb. 164673
- Morel-Journel, T., V. Thuillier, F. Pennekamp, E. Laurent, D. Legrand, A. S. Chaine, and N. Schtickzelle. 2020. A multidimensional approach to the expression of phenotypic plasticity. Funct. Ecol. **34**: 2338–2349. doi:10.1111/1365-2435.13667
- Morón Lugo, S. C., M. Baumeister, O. M. Nour, F. Wolf, M. Stumpp, and C. Pansch. 2020. Warming and temperature variability determine the performance of two invertebrate predators. Sci. Rep. **10**: 6780. doi:10.1038/s41598-020-63679-0
- Nejstgaard, J. C., and others. 2006. Plankton development and trophic transfer in seawater enclosures with nutrients and *Phaeocystis pouchetii* added. Mar. Ecol. Prog. Ser. **321**: 99–121. doi:10.3354/meps321099
- Nejstgaard, J. C., K. W. Tang, M. Steinke, J. Dutz, M. Koski, E. Antajan, and J. D. Long. 2007. Zooplankton grazing on *Phaeocystis*: A quantitative review and future challenges. Biogeochemistry 83: 147–172. doi:10.1007/s10533-007-9098-y
- Neun, S., N. H. Hintz, M. Schröder, and M. Striebel. 2022. Phytoplankton response to different light colors and fluctuation frequencies. Front. Mar. Sci. **9**: 824624. doi:10.3389/ fmars.2022.824624
- Niehaus, A. C., M. J. Angilletta Jr., M. W. Sears, C. E. Franklin, and R. S. Wilson. 2012. Predicting the physiological performance of ectotherms in fluctuating thermal environments. J. Exp. Biol. **215**: 694–701. doi:10.1242/jeb.058032
- Oliver, E. C., and others. 2018. Longer and more frequent marine heatwaves over the past century. Nat. Commun. **9**: 1324. doi:10.1038/s41467-018-03732-9
- Pansch, C., and others. 2018. Heat waves and their significance for a temperate benthic community: A near-natural experimental approach. Glob. Change Biol. **24**: 4357– 4367. doi:10.1111/gcb.14282
- Pansch, C., and C. Hiebenthal. 2019. A new mesocosm system to study the effects of environmental variability on marine species and communities. Limnol. Oceanogr.: Methods **17**: 145–162. doi:10.1002/lom3.10306
- Pansch, C., M. Raatz, S. Comeau, T. T. Hui, J. N. Havenhand, J. Vajedsamiei, and C. E. Cornwall. 2022. Influence of

environmental variability on climate change impacts in marine ecosystems. Front. Mar. Sci. **9**: 994756. doi:10. 3389/fmars.2022.994756

- Peck, L. S., S. A. Morley, J. Richard, and M. S. Clark. 2014. Acclimation and thermal tolerance in Antarctic marine ectotherms. J. Exp. Biol. **217**: 16–22. doi:10.1242/jeb. 089946
- Peteiro, L. G., S. A. Woodin, D. S. Wethey, D. Costas-Costas, A. Martínez-Casal, C. Olabarria, and E. Vázquez. 2018. Responses to salinity stress in bivalves: Evidence of ontogenetic changes in energetic physiology on *Cerastoderma edule*. Sci. Rep. 8: 8329. doi:10.1038/s41598-018-26706-9
- Petersen, J. E., V. S. Kennedy, W. C. Dennison, and W. M. Kemp [eds.]. 2009, *Enclosed experimental ecosystems and scale: Tools for understanding and managing coastal ecosystems*. Springer.
- Picoche, C., and F. Barraquand. 2020. Strong self-regulation and widespread facilitative interactions in phytoplankton communities. J. Ecol. **108**: 2232–2242. doi:10.1111/1365-2745.13410
- Polazzo, F., S. K. Roth, M. Hermann, A. Mangold-Döring, A. Rico, A. Sobek, P. J. Van den Brink, and M. C. Jackson. 2022. Combined effects of heatwaves and micropollutants on freshwater ecosystems: Towards an integrated assessment of extreme events in multiple stressors research. Glob. Change Biol. 28: 1248–1267. doi:10.1111/gcb.15971
- Pörtner, H.-O. 2010. Oxygen-and capacity-limitation of thermal tolerance: A matrix for integrating climate-related stressor effects in marine ecosystems. J. Exp. Biol. 213: 881–893. doi:10.1242/jeb.037523
- Quévreux, P., M. Barbier, and M. Loreau. 2021. Synchrony and perturbation transmission in trophic metacommunities. Am. Nat. **197**: E188–E203. doi:10.1086/714131
- Raatz, M., E. van Velzen, and U. Gaedke. 2019. Co-adaptation impacts the robustness of predator–prey dynamics against perturbations. Ecol. Evol. 9: 3823–3836. doi:10.1002/ece3.5006
- Rasconi, S., K. Winter, and M. J. Kainz. 2017. Temperature increase and fluctuation induce phytoplankton biodiversity loss—evidence from a multi-seasonal mesocosm experiment. Ecol. Evol. 7: 2936–2946. doi:10.1002/ece3.2889
- Rasmusson, L. M., P. Buapet, R. George, M. Gullström, P. C. Gunnarsson, and M. Björk. 2020. Effects of temperature and hypoxia on respiration, photorespiration, and photosynthesis of seagrass leaves from contrasting temperature regimes. ICES J. Mar. Sci. **77**: 2056–2065. doi:10.1093/ icesjms/fsaa093
- Rescan, M., D. Grulois, E. Ortega-Aboud, and L.-M. Chevin. 2020. Phenotypic memory drives population growth and extinction risk in a noisy environment. Nat. Ecol. Evol. 4: 193–201. doi:10.1038/s41559-019-1089-6
- Rezende, E. L., L. E. Castañeda, and M. Santos. 2014. Tolerance landscapes in thermal ecology. Funct. Ecol. **28**: 799– 809. doi:10.1111/1365-2435.12268

23782242, 0, Downloaded from

https://aslopubs

- Rezende, E. L., and F. Bozinovic. 2019. Thermal performance across levels of biological organization. Philos. Trans.
 R. Soc. Lond. B Biol. Sci. 374: 20180549. doi:10.1098/rstb. 2018.0549
- Richmond, C. E., and S. A. Woodin. 1996. Short-term fluctuations in salinity: Effects on planktonic invertebrate larvae. Mar. Ecol. Prog. Ser. 133: 167–177. doi:10.3354/meps133167
- Rivest, E. B., S. Comeau, and C. E. Cornwall. 2017. The role of natural variability in shaping the response of coral reef organisms to climate change. Curr. Clim. Change Rep. 3: 271–281. doi:10.1007/s40641-017-0082-x
- Roman, M. R., S. B. Brandt, E. D. Houde, and J. J. Pierson. 2019. Interactive effects of hypoxia and temperature on coastal pelagic zooplankton and fish. Front. Mar. Sci. 6: 139. doi:10.3389/fmars.2019.00139
- Ruel, J. J., and M. P. Ayres. 1999. Jensen's inequality predicts effects of environmental variation. Trends Ecol. Evol. **14**: 361–366. doi:10.1016/S0169-5347(99)01664-X
- Ryo, M., C. A. Aguilar-Trigueros, L. Pinek, L. A. Muller, and M. C. Rillig. 2019. Basic principles of temporal dynamics. Trends Ecol. Evol. **34**: 723–733. doi:10.1016/j.tree.2019. 03.007
- Safaie, A., and others. 2018. High frequency temperature variability reduces the risk of coral bleaching. Nat. Commun. 9: 1671. doi:10.1038/s41467-018-04074-2
- Sanford, E., and M. W. Kelly. 2011. Local adaptation in marine invertebrates. Ann. Rev. Mar. Sci. 3: 509–535. doi: 10.1146/annurev-marine-120709-142756
- Sawall, Y., M. Ito, and C. Pansch. 2021. Chronically elevated sea surface temperatures revealed high susceptibility of the eelgrass *Zostera marina* to winter and spring warming. Limnol. Oceanogr. 66: 4112–4124. doi:10.1002/lno.11947
- Schwager, M., K. Johst, and F. Jeltsch. 2006. Does red noise increase or decrease extinction risk? Single extreme events versus series of unfavorable conditions. Am. Nat. 167: 879– 888. doi:10.1086/503609
- Shatwell, T., A. Nicklisch, and J. Köhler. 2012. Temperature and photoperiod effects on phytoplankton growing under simulated mixed layer light fluctuations. Limnol. Oceanogr. 57: 541–553. doi:10.4319/lo.2012.57.2.0541
- Shama, L. N. 2017. The mean and variance of climate change in the oceans: hidden evolutionary potential under stochastic environmental variability in marine sticklebacks. Sci. Rep. **7**: 1–14.
- Simon, F. W., and D. A. Vasseur. 2021. Variation cascades: Resource pulses and top-down effects across time scales. Ecology **102**: e03277. doi:10.1002/ecy.3277
- Sperfeld, E., D. Raubenheimer, and A. Wacker. 2016. Bridging factorial and gradient concepts of resource co-limitation: Towards a general framework applied to consumers. Ecol. Lett. **19**: 201–215. doi:10.1111/ele.12554
- Strauss, S. Y. 2014. Ecological and evolutionary responses in complex communities: Implications for invasions and eco-

evolutionary feedbacks. Oikos **123**: 257–266. doi:10.1111/j.1600-0706.2013.01093.x

- Striebel, M., S. Schabhüttl, D. Hodapp, P. Hingsamer, and H. Hillebrand. 2016. Phytoplankton responses to temperature increases are constrained by abiotic conditions and community composition. Oecologia 182: 815–827. doi:10. 1007/s00442-016-3693-3
- Theus, M. E., T. J. Layden, N. McWilliams, S. Crafton-Tempel, C. T. Kremer, and S. B. Fey. 2022. Photoperiod influences the shape and scaling of freshwater phytoplankton responses to light and temperature. Oikos **2022**: e08839. doi:10.1111/oik.08839
- Thomas, M. K., M. Aranguren-Gassis, C. T. Kremer, M. R. Gould, K. Anderson, C. A. Klausmeier, and E. Litchman. 2017. Temperature-nutrient interactions exacerbate sensitivity to warming in phytoplankton. Glob. Change Biol. 23: 3269–3280. doi:10.1111/gcb.13641
- Thompson, R. M., J. Beardall, J. Beringer, M. Grace, and P. Sardina. 2013. Means and extremes: Building variability into community-level climate change experiments. Ecol. Lett. **16**: 799–806. doi:10.1111/ele.12095
- Thomsen, M., T. Wernberg, J. Olden, J. E. Byers, J. Bruno, B. Silliman, and D. Schiel. 2014. Forty years of experiments on aquatic invasive species: Are study biases limiting our understanding of impacts? NeoBiota **22**: 1–22. doi:10. 3897/neobiota.22.6224
- Travis, J. M. 2001. The color of noise and the evolution of dispersal. Ecol. Res. **16**: 157–163. doi:10.1046/j.1440-1703. 2001.00381.x
- Urrutia-Cordero, P., and others. 2021. SITES AquaNet: An open infrastructure for mesocosm experiments with high frequency sensor monitoring across lakes. Limnol. Oceanogr.: Methods **19**: 385–400. doi:10.1002/lom3.10432
- Vajedsamiei, J., F. Melzner, M. Raatz, R. Kiko, M. Khosravi, and C. Pansch. 2021a. Simultaneous recording of filtration and respiration in marine organisms in response to shortterm environmental variability. Limnol. Oceanogr.: Methods 19: 196–209. doi:10.1002/lom3.10414
- Vajedsamiei, J., F. Melzner, M. Raatz, S. C. Morón Lugo, and C. Pansch. 2021b. Cyclic thermal fluctuations can be burden or relief for an ectotherm depending on fluctuations' average and amplitude. Funct. Ecol. 35: 2483–2496. doi:10. 1111/1365-2435.13889
- Vajedsamiei, J., M. Wahl, A. L. Schmidt, M. Yazdanpanahan, and C. Pansch. 2021*c*. The higher the needs, the lower the tolerance: Extreme events may select ectotherm recruits with lower metabolic demand and heat sensitivity. Front. Mar. Sci. **8**: 660427. doi:10. 3389/fmars.2021.660427
- Vaquer-Sunyer, R., and C. M. Duarte. 2011. Temperature effects on oxygen thresholds for hypoxia in marine benthic organisms. Glob. Change Biol. **17**: 1788–1797. doi:10. 1111/j.1365-2486.2010.02343.x

23782242, 0, Downloaded from

https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/tol2.10286 by HGF GEOMAR Helmholtz Centre of Ocean, Wiley Online Library on [01/11/2022]. See the Terms

and Conditions

(https://onlinelibrary.wiley.com

conditions) on Wiley Online I

library for rules of use; OA articles are

governed by the applicable Creative Commons

- Vasseur, D. A., and P. Yodzis. 2004. The color of environmental noise. Ecology **85**: 1146–1152. doi:10.1890/02-3122
- Vasseur, D. A., and others. 2014. Increased temperature variation poses a greater risk to species than climate warming. Proc. R. Soc. B: Biol. Sci. 281: 20132612. doi:10.1098/rspb. 2013.2612
- Verbeek, L., A. Gall, H. Hillebrand, and M. Striebel. 2018. Warming and oligotrophication cause shifts in freshwater phytoplankton communities. Glob. Change Biol. 24: 4532–4543. doi:10.1111/gcb.14337
- Vinebrooke, R., K. L. Cottingham, M. S. Norberg, S. I. Dodson, S. C. Maberly, and U. Sommer. 2004. Impacts of multiple stressors on biodiversity and ecosystem functioning: The role of species co-tolerance. Oikos **104**: 451–457. doi:10. 1111/j.0030-1299.2004.13255.x
- Wahl, M., V. Saderne, and Y. Sawall. 2016. How good are we at assessing the impact of ocean acidification in coastal systems? Limitations, omissions and strengths of commonly used experimental approaches with special emphasis on the neglected role of fluctuations. Mar. Freshw. Res. **67**: 25–36. doi:10.1071/MF14154
- Wahl, M., S. Schneider Covachã, V. Saderne, C. Hiebenthal, J. D. Müller, C. Pansch, and Y. Sawall. 2018. Macroalgae may mitigate ocean acidification effects on mussel calcification by increasing pH and its fluctuations. Limnol. Oceanogr. 63: 3–21. doi:10.1002/lno.10608
- Wahl, M., and others. 2021. Pulsed pressure: Fluctuating impacts of multifactorial environmental change on a temperate macroalgal community. Limnol. Oceanogr. 66: 4210–4226. doi:10.1002/lno.11954
- Walters, R. J., W. U. Blanckenhorn, and D. Berger. 2012. Forecasting extinction risk of ectotherms under climate warming: An evolutionary perspective. Funct. Ecol. 26: 1324–1338. doi:10.1111/j.1365-2435.2012.02045.x
- Wang, G., and M. E. Dillon. 2014. Recent geographic convergence in diurnal and annual temperature cycling flattens global thermal profiles. Nat. Clim. Change 4: 988–992. doi: 10.1038/nclimate2378
- Wang, X., F. Fu, P. Qu, J. D. Kling, H. Jiang, Y. Gao, and D. A. Hutchins. 2019. How will the key marine calcifier *Emiliania huxleyi* respond to a warmer and more thermally variable ocean? Biogeosciences **16**: 4393–4409. doi:10.5194/bg-16-4393-2019
- Wesselmann, M., R. M. Chefaoui, N. Marba, E. Serrao, and C. M. Duarte. 2021. Warming threatens to propel the expansion of the exotic seagrass *Halophila stipulacea*. Front. Mar. Sci 8: 759676. doi:10.3389/fmars.2021.759676
- Wolf, F., K. Seebass, and C. Pansch. 2022. The role of recovery phases in mitigating the negative impacts of marine heatwaves on the sea star Asterias rubens. Front. Mar. Sci. 8: 790241.
- Woolway, R. I., E. Jennings, T. Shatwell, M. Golub, D. C. Pierson, and S. C. Maberly. 2021. Lake heatwaves under

climate change. Nature **589**: 402–407. doi:10.1038/s41586-020-03119-1

- Yvon-Durocher, G., J. I. Jones, M. Trimmer, G. Woodward, and J. M. Montoya. 2010. Warming alters the metabolic balance of ecosystems. Philos. Trans. R. Soc. Lond. B Biol. Sci. 365: 2117–2126. doi:10.1098/rstb.2010.0038
- Yvon-Durocher, G., C.-E. Schaum, and M. Trimmer. 2017. The temperature dependence of phytoplankton stoichiometry: Investigating the roles of species sorting and local adaptation. Front. Microbiol. 8: 2003. doi:10.3389/fmicb. 2017.02003
- Zhang, M., B. Qin, Y. Yu, Z. Yang, X. Shi, and F. Kong. 2016. Effects of temperature fluctuation on the development of cyanobacterial dominance in spring: Implication of future climate change. Hydrobiologia **763**: 135–146. doi:10.1007/ s10750-015-2368-0
- Zhang, M., Y. Guan, B. Qin, and X. Wang. 2019. Responses of phytoplankton species to diel temperature fluctuation patterns. Phycol. Res. 67: 184–191. doi:10.1111/pre.12369
- Ziegler, M., and others. 2021. Integrating environmental variability to broaden the research on coral responses to future ocean conditions. Glob. Change Biol. **27**: 5532–5546. doi: 10.1111/gcb.15840
- Zurell, D., F. Jeltsch, C. F. Dormann, and B. Schröder. 2009. Static species distribution models in dynamically changing systems: How good can predictions really be? Ecography **32**: 733–744. doi:10.1111/j.1600-0587.2009. 05810.x

Acknowledgments

This project has received funding from the European Union Horizon 2020 research and innovation program AQUACOSM-plus under grant agreement No 871081. MG, DCM, JCN, SAB, MB, KAG, SL were supported by INFRAIA EU-H2020 AQUACOSM-plus and MB by EU-H2020, PONDERFUL (No 869296) project. MS and PT were supported by the German Research Foundation DFG STR 1383/8-1. MCP was supported by the Irish Marine Institute funded under the Marine Research Program by the Irish Government named BEYOND2020 (Grant-Aid Agreement No. PBA/FS/16/02) and the AQUACOSM TA grant (2019) under the project name AlloEcoMetry. FP was supported by H2020-MSCA-ITN ECORISK2050 project (Grant Agreement nr. 813124). SBF was supported by NSF DEB 1856415. RMC was supported by MARHOT project from the EU H2020 research and innovation programme under the Marie Skłodowska-Curie Grant N° 894941. JV and CP were supported by the German Research Foundation DFG (Grant Number: PA 2643/2/348431475). Open Access funding enabled and organized by Projekt DEAL. WOA Institution: CARL VON OSSIETZKY UNIVERSITAET OLDENBURG Consortia Name : Projekt DEAL

> Submitted 18 February 2022 Revised 14 September 2022 Accepted 28 September 2022